Author's response to reviews

Title: The assessment of the prognosis of musculoskeletal conditions in older adults presenting to general practice: a research protocol

Authors:

Christian D Mallen (c.d.mallen@cphc.keele.ac.uk)
George Peat (g.m.peat@cphc.keele.ac.uk)
Elaine Thomas (e.thomas@cphc.keele.ac.uk)
Simon Wathall (s.wathall@cphc.keele.ac.uk)
Tracy Whitehurst (t.whitehurst@cphc.keele.ac.uk)
Charlotte Clements (c.v.clements@cphc.keele.ac.uk)
Joanne Bailey (j.bailey@cphc.keele.ac.uk)
Jacqueline Gray (j.gray@cphc.keele.ac.uk)
Peter R Croft (p.r.croft@cphc.keele.ac.uk)

Version: 2 Date: 26 October 2006

Author's response to reviews: see over
Primary Care Sciences Research Centre
Keele University
Keele
Staffordshire
ST5 3BG
26th October 2006

Dear Editor,

Re: MS: 1783443396116875

Thank you for your decision to publish the study protocol for the PROG-RES study in BMC Musculoskeletal Disorders. I have added a conflict of interest’s statement as requested.

Should you require any further information, please do not hesitate to contact me.

Yours sincerely

Dr Christian Mallen
arc Primary Care Research Fellow
Email: c.d.mallen@cphc.keele.ac.uk